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Abstract 
 

The abundance of natural gas reserves and tremendous progress in its production has drawn great 

interest from the research community to develop a suitable technology for the utilization of stranded 

natural gas, into transportable chemicals/fuels which are of great commercial value. Catalytic 

transformation of the simplest hydrocarbon methane into easily transportable hydrocarbons such as 

benzene and aromatics in oxygen-free conditions is of significant importance that affords an alternate 

path for the utilization of methane to fulfill the growing energy demand. Mo-exchanged zeolite (ZSM-

5) is an extensively recognized catalyst for this reaction that behaves in a bifunctional manner providing 

active sites for methane activation and benzene formation. The major challenge associated with 

thermodynamic restraint methane dehydroaromatization (MDA) using molybdenum-impregnated 

HZSM-5 zeolite  (Mo/HZSM-5) catalyst is the generation of an extensive amount of coke inside the 

sole micropores and over the surface which leads to the deactivation of the catalyst within a few hours. 

These problems require more insights into the development of suitable MDA catalysts with improved 

catalytic activity in view of upgrading the process to a commercial level. In this context, the present 

research covers the introduction of hierarchical zeolite comprised of unique micro/mesoporous 

structures that can assist in delivering homogeneous Mo dispersion while the additional pores 

accelerate the mass transfer of reactant and formed benzene products. With this approach, the first 

study deals with the synthesis of hierarchical ZSM-5 support via applying a dual template soft 

templating strategy and its application for the MDA reaction. Another factor that affects the MDA 

catalytic activity is the dispersibility and stability of active molybdenum species under the reaction 

condition. A strategy to stabilize the Mo-oxide precursor over the zeolite support is proposed and, in 

this regard, two different forms of MoO3 (α-MoO3, h-MoO3) were synthesized and applied as Mo 

precursors for the synthesis of microwave (MW) assisted Mo/HZSM-5 catalysts. The correspondence 
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between the impact of different Mo oxide precursors and their preparation method on the catalytic 

activity of Mo/ZHSM-5 was established for the non-oxidative MDA process that provides maximum 

benzene selectivity with affordable stability. In order to reduce the coke generation inside the zeolite 

support, the Bronsted acidity of HZSM-5 zeolite is moderated by substituting trivalent Al atom with 

various heteroatoms (V, B, Ga, & Fe). Modification in Bronsted acidic sites of zeolite in Mo/MFI 

catalyst showed improved stability towards MDA by reducing the coke deposition thus extending the 

catalyst lifetime. In parallel efforts, ceria-zirconia mixed oxides as an oxygen transfer agent along with 

Mo/HZSM-5 to improve the catalytic stability by performing the in-situ hydrogen oxidation reaction 

was accomplished which in turn improved the methane conversion (11% initially) and benzene 

selectivity (70%) at 90-120 minutes of MDA reaction.  Additionally, the promotional role of Ag and 

its addition with Mo/HZSM-5 catalysts were explored to improve the MDA catalytic activity and 

stability. Investigation revealed that the co-impregnation of Ag with Mo/HZSM-5 significantly 

increases the total acidity (7.8%) and hence aromatization ability of the catalyst thereby providing a 

higher benzene formation rate (320 nmol/gcat. s), with a decreased amount of coke generation over the 

catalyst surface. Performed characterization showed that Ag addition modified the chemical 

environment of Mo/HZSM-5 which possesses superior catalytic activity towards MDA. Overall, this 

dissertation focused on developing a suitable MDA catalyst via modifying the support porosity, acidity, 

Mo precursor's state, and promoter's effect to reduce the coke formation with simultaneous 

improvement in benzene productivity toward methane aromatization. 
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सार 
प्राकृतिक गैस भंडार की प्रचुरिा और इसके उत्पादन में जबरदस्त प्रगति ने फंसे हुए प्राकृतिक गैस के पररवहन 

योग्य रसायनो/ंईंधन में उपयोग के तिए एक उपयुक्त िकनीक तवकतसि करने के तिए अनुसंधान समुदाय में 

काफी रुतच पैदा की है, जो तक बडे वातिज्यिक मूल्य के हैं। सरििम हाइडर ोकाबबन मीथेन का आसानी से 

पररवहन योग्य हाइडर ोकाबबन जैसे बेंजीन और एरोमैतिक्स में ऑक्सीजन मुक्त ज्यितियो ंमें उते्प्ररक पररविबन 

महत्वपूिब महत्व रखिा है जो बढ़िी ऊजाब मांग को पूरा करने के तिए मीथेन के उपयोग के तिए एक वैकज्यिक 

मागब प्रदान करिा है। मो-एक्सचेंज्ड तजओिाइि (ZSM-5) इस प्रतितिया के तिए एक व्यापक रूप से मान्यिा 

प्राप्त उते्प्ररक है जो मीथेन सतियि और बेंजीन गठन के तिए सतिय साइि प्रदान करने वािे ति-कायाबत्मक 

िरीके से व्यवहार करिा है। मोतिब्डेनम-संसेतचि HZSM-5 तजओिाइि (Mo/HZSM-5) उते्प्ररक का उपयोग 

करके थमोडायनातमक संयम मीथेन डीहाइडर ोएरोमेिाइजेशन (एमडीए) से जुडी प्रमुख चुनौिी एकमात्र 

माइिोप्रोसब के अंदर और सिह पर व्यापक मात्रा में कोक का उत्पादन है जो तनज्यियिा की ओर िे जािी है। 

कुछ ही घंिो ंमें उते्प्ररक का. इन समस्याओ ंके तिए प्रतिया को व्यावसातयक स्तर पर अपगे्रड करने के मदे्दनजर 

बेहिर उते्प्ररक गतितवतध के साथ उपयुक्त एमडीए उते्प्ररक के तवकास में अतधक अंिर्दबति की आवश्यकिा है। 

इस संदभब में, विबमान शोध में अतििीय सूक्ष्म/मेसोपोरस संरचनाओ ं से युक्त पदानुितमि तजओिाइि की 

शुरूआि को शातमि तकया गया है जो सजािीय मो फैिाव प्रदान करने में सहायिा कर सकिा है जबतक 

अतिररक्त तछद्र प्रतितियाशीि और गतठि बेंजीन उत्पादो ंके बडे पैमाने पर िानांिरि में िेजी िािे हैं। इस 

र्दतिकोि के साथ, पहिा अध्ययन दोहरे िेम्पिेि सॉफ्ट िेम्पिेतिंग रिनीति को िागू करने और एमडीए 

प्रतितिया के तिए इसके अनुप्रयोग के माध्यम से पदानुितमि ZSM-5 समथबन के संशे्लषि से संबंतधि है। एक 

अन्य कारक जो एमडीए उते्प्ररक गतितवतध को प्रभातवि करिा है वह प्रतितिया ज्यिति के िहि सतिय 

मोतिब्डेनम प्रजातियो ंकी फैिावशीििा और ज्यिरिा है। तजओिाइि समथबन पर मो-ऑक्साइड अग्रदूि को 

ज्यिर करने की एक रिनीति प्रस्तातवि है और, इस संबंध में, MoO3 (α-MoO3, h-MoO3) के दो अिग-

अिग रूपो ंको संशे्लतषि तकया गया और माइिोवेव (MW) के संशे्लषि के तिए Mo अग्रदूिो ंके रूप में िागू 

तकया तवतभन्न Mo ऑक्साइड अग्रदूिो ंके प्रभाव और Mo/ZHSM-5 की उते्प्ररक गतितवतध पर उनकी िैयारी 

तवतध के बीच पत्राचार गैर-ऑक्सीडेतिव एमडीए प्रतिया के तिए िातपि तकया गया था जो तकफायिी ज्यिरिा 

के साथ अतधकिम बेंजीन चयनात्मकिा प्रदान करिा है। तजओिाइि समथबन के अंदर कोक उत्पादन को कम 

करने के तिए, HZSM-5 तजओिाइि की ब्ोसें्टड अम्लिा को तत्रसंयोजक अि परमािु को तवतभन्न हेिेरोएिम 

(वी, बी, गा, और फे) के साथ प्रतििातपि करके तनयंतत्रि तकया जािा है। एमओ/एमएफआई उते्प्ररक में 
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तजओिाइि के ब्ोसें्टड अम्लीय साइिो ंमें संशोधन ने कोक जमाव को कम करके एमडीए के प्रति बेहिर ज्यिरिा 

तदखाई, तजससे उते्प्ररक का जीवनकाि बढ़ गया। समानांिर प्रयासो ं में, सेररया-त़िरकोतनया ने इन-सीिू 

हाइडर ोजन ऑक्सीकरि प्रतितिया करके उते्प्ररक ज्यिरिा में सुधार करने के तिए Mo/HZSM-5 के साथ 

ऑक्सीजन िर ांसफर एजेंि के रूप में ऑक्साइड को तमतिि तकया, तजसके पररिामस्वरूप मीथेन रूपांिरि में 

सुधार हुआ (शुरुआि में 11%) और एमडीए प्रतितिया के 90-120 तमनि पर बेंजीन चयनात्मकिा (70%)। 

इसके अतिररक्त, एमडीए उते्प्ररक गतितवतध और ज्यिरिा में सुधार के तिए एजी की प्रचार भूतमका और 

एमओ/एचजेडएसएम-5 उते्प्ररक के साथ इसके जुडाव का पिा िगाया गया। जांच से पिा चिा तक 

Mo/HZSM-5 के साथ Ag के सह-संसेचन से कुि अम्लिा (7.8%) काफी बढ़ जािी है और इसतिए उते्प्ररक 

की सुगंधीकरि क्षमिा कम हो जािी है, तजससे उच्च बेंजीन तनमाबि दर (320 nmol/gcat. s) तमििी है। 

उते्प्ररक सिह पर कोक उत्पादन की मात्रा। प्रदशबन तकए गए िक्षि विबन से पिा चिा तक एजी के अतिररक्त 

ने Mo/HZSM-5 के रासायतनक वािावरि को संशोतधि तकया है तजसमें एमडीए के प्रति बेहिर उते्प्ररक 

गतितवतध है। कुि तमिाकर, यह शोध प्रबंध मीथेन सुगंधीकरि की तदशा में बेंजीन उत्पादकिा में एक साथ 

सुधार के साथ कोक गठन को कम करने के तिए समथबन सरंध्रिा, अम्लिा, मो अग्रदूि की ज्यिति और प्रमोिर 

के प्रभाव को संशोतधि करके एक उपयुक्त एमडीए उते्प्ररक तवकतसि करने पर कें तद्रि है। 
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